Author's response to reviews

Title: Effects of iodinated contrast agents on renal oxygenation level by BOLD-MRI in a rabbit model of diabetic nephropathy

Authors:

Jia-huan Wang (wangjiahuan.2008@163.com)
Ke Ren (zhongguoyidarenke@163.com)
Wen-ge Sun (wengesun_zgyd1@163.com)
Li Zhao (m15840597875@163.com)
Hong-shan Zhong (wangjiahuan_2008@yeah.net)
Ke Xu (zhongguoyidaxuke@126.com)

Version: 2
Date: 18 June 2014

Author's response to reviews: see over
Dear Editor,

We would like to submit a research article, “Effects of iodinated contrast agents on renal oxygenation level by BOLD-MRI in a rabbit model of diabetic nephropathy”, by jia-huan wang, et al for consideration for publication in BMC Nephrology.

This paper has not been published elsewhere in whole or in part. All authors have read and approved the content, and agree to submit it for consideration for publication in your journal. There are no ethical/legal conflicts involved in the article.

Your consideration for this manuscript is highly appreciated.

We look forward to hearing from you soon.

Yours sincerely,

Jia-huan wang

Corresponding authors
Dr. Ke Ren  Phone: +86-15840597875, Email: zhongguoyidarenke@163.com
Dr. Ke Xu  Phone:+86-13596221566, Email: zhongguoyidaxuke@126.com